Author’s response to reviews

Title: A case of osteosarcoma in the distal femur two years after an ipsilateral femoral shaft fracture. Coincidence? A case report

Authors:

Oscar Dorrestijn (o.dorrestijn@orth.umcg.nl)
Paul C Jutte (p.c.jutte@orth.umcg.nl)

Version: 2 Date: 14 April 2010

Author’s response to reviews:

Dear editor,

Enclosed is the revised version of our manuscript “A case of osteosarcoma in the distal femur two years after an ipsilateral femoral shaft fracture. Coincidence? A case report.”, submitted for publication in the Journal of Medical Case Reports.

The following format changes have been made, as suggested:

• Our abstract is restructured as proposed.
• The ethnicity of the patient is added.
• A conclusion section is included.
• A figure legend section after the reference list is included.
• The figures are provided without legend.

Awaiting your reaction, on behalf of the co-author.

Yours sincerely,

Oscar Dorrestijn